RECORDING REQUESTED BY AND WHEN RECORDED RETURN TO:

RIVER BEND LIMITED PARTNERSHIP 15 S.W. COLORADO, SUITE A BEND, OREGON 97702

98-27427

MARY SUE PENHOLLOW, COUNTY CORDER OF CONVEYANCES, IN AND THE THAT STRUMENT WAS RECORDED THIS DAY 98 JUN 26 AM 9: 27

- 2948

AMENDED AND RESTATED
MASTER DECLARATION OF COVENANTS,
CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

STATE OF OREGON

# RIVER BEND LIMITED PARTNERSHIP

("Declarant")

Dated Effective as of November 4, 1997

# TABLE OF CONTENTS

			Page	
1.	Definitions			
	1.1	Certain Defined Terms.	 1	
	1.2	Liberal Construction.	1	
	1.3	Interpretation	0	
	1.4	Amendment and Restatement.	6	
2.	Anne	Annexation and Withdrawal of Property		
	2.1	Annexation.	7	
	2.2	Withdrawal of Property.	/	
	2.3	Assignment of Rights	8	
3.	Prope	Property Rights.		
	3.1	Common Area		
	3.2	Limited Common Area.	 م	
	3.3	Delegation.	9	
4.	Restri	ctions on Use.		
	4.1	General Applicability.	9	
	4.2	Declarant Authority.	9	
	4.3	Board Authority.	9	
	4.4	Members' Power.	10	
	4.5	Owners' Acknowledgment.	10	
	4.6	General Restrictions.	10	
	4.7	Limitations.	10	
	4.8	Development.	10 11	
5.	Easem	ents.		
	5.1	Easements for Encroachments.	.11	
	5.2	Easements for Utilities, Etc.	.11	
	5.3	Easements for Construction and Sales Activities.	.12	
	5.4	Easements for Drainage	.13	
	5.5	Easements for Drainage.	.13	
	5.6	Right of Entry.  Easements for Vehicular Ingress, Egress and Parking	.13	
	5.7	Easements Deemed Created	.14 .14	
6.	Design	Review Committee		
	6.1	Design Review	.14	
	6.2	Design Guidelines.	15	
	6.3	Governmental Approvals.	15	
	6.4	variance	1.5	
	6.5	Committee Decision	15	

	6.6	Committee Discretion.	15				
	6.7	Membership; Appointment and Removal	16				
	6.8	Majority Action	16				
	6.9	Limitation of Liability.	16				
	6.10	Nonwaiver					
	6.11	Effective Period of Consent.					
	6.12	Estoppel Certificate.					
	6.13	Activities of Declarant.					
7.	Const	Construction and Alteration of Improvements; Sewer Services and					
	Irrigation Water						
	7.1	General	17				
	7.2	Construction Methods	17				
	7.3	Completion of Improvements.	18				
	7.4	Setback, Maximum Height, and Minimum Yard Requirements	18				
	7.5	Sewer Services and Irrigation Water.					
8.	Restrictions on Partition and Combination.						
	8.1	Parcels	18				
	8.2	Common Area	19				
9.	Association Function, Membership, and Voting Rights						
	9.1	Function of Association.					
	9.2	Membership.	19				
	9.3	Voting	19				
	9.4	Exercise of Voting Rights.	20				
	9.5	Turnover					
10.	Rights and Obligations of the Association						
	10.1	Area of Common Responsibility.	21				
	10.2	Real and Personal Property for Common Use.	21				
	10.3	Enforcement	21				
	10.4	Implied Rights; Board Authority.	21				
	10.5	Property for Public Use	21				
	10.6	Professional Management; Contracts.	22				
	10.7	Liability and Indemnification.	22				
	10.8	Security.	22				
11.	Assessments.						
	11.1	Creation of Assessments.					
	11.2	Determination of Base Assessment.	24				
	11.3	Reserves.					
	11.4	Special Assessments.	25				
	11.5	Specific Assessments					
	11.6	Limitation on Increases in Base Assessments.					
	11.7	Date of Commencement of Assessments	26				

	11.8	Lien for Assessments.	26	
	11.9	Failure to Assess.	27	
	11.10	Exempt Property.	27	
	11.11	Reallocation Upon Annexation or Withdrawal of Property	27	
	11.12			
	11.13			
12.	Enforcement of Covenants			
	12.1	Remedies	28	
	12.2	Rights of Owners.	29	
	12.3	Limitations on Liability.		
	12.4	Recovery of Costs and Fees.		
	12.5	Remedies Not Exclusive.		
13.	Mainte	enance of Properties.	30	
	13.1	Association's Responsibility		
	13.2	Owner's Responsibility		
14.	Insurance and Casualty Losses.			
	14.1	Association Insurance.	31	
	14.2	Owners' Insurance.	32	
	14.3	Damage and Destruction.	33	
	14.4	Association as Attorney-in-Fact.	34	
15.	Condemnation			
	15.1	Partial Condemnation	34	
	15.2	Complete Condemnation.	34	
16.	Mortg	Mortgage Provisions.		
	16.1	Parties Benefited	35	
	16.2	Notices.	35	
	16.3	Special FNMA/FMC Provision.	35	
	16.4	No Priority.	36	
	16.5	Notice to Association	36	
	16.6	Amendment by Board	36	
	16.7	Failure of Mortgagee to Respond.	36	
17.	Declarant's Rights			
	17.1	Transfer of Rights.		
	17.2	Construction and Sales Period.		
	17.3	Other Covenants		
	17.4	Amendment		
	17.5	Termination of Special Rights.	37	
18.		ral Provisions.		
	18.1	Term	37	

# 499 - 2952

		38
18.2	Amendment	39
18.3	Severability Notice of Sale or Transfer of Title	39
18.4	Notice of Sale or Transfer of Title	39
18.5	Revocation.	39
18.6	Notices	30
18.7	Waiver	40
18.8	Conflicts Between Documents	**************************************

# AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

This AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND (this "Declaration") is made effective as of the 4th day of November, 1997, by RIVER BEND LIMITED PARTNERSHIP, an Oregon limited partnership ("Declarant").

Declarant is the owner of the real property in Deschutes County, Oregon, legally described on the attached Exhibit A. Declarant intends to cause the Properties (defined in Section 1.1) to be developed as a multi-phase mixed-use project containing commercial, industrial, and residential components, as well as common areas. In connection therewith, Declarant recorded that certain Master Declaration of Covenants, Conditions, and Restrictions for River Bend in the Official Records of Deschutes County, Oregon on November 4, 1997 in Book 468, Page 1683, Document No. 97-41096 (the "Original Declaration") and Declarant desires to amend and restate the Original Declaration in its entirety as provided herein.

This Declaration imposes, and the Supplemental Declarations (defined in Section 1.1) contemplated herein will impose, upon the Properties mutually-beneficial restrictions under a general plan of improvement for the benefit of the owners of each portion of the Properties and establishes procedures for the overall development, administration, maintenance, and preservation of the Properties. Through this Declaration, Declarant seeks to ensure the attractiveness of the individual lots, parcels, and facilities developed within the Properties and to preserve, protect, and enhance the value and amenities of the Properties.

Declarant hereby declares that the Properties and all parts thereof shall be held, used, mortgaged, and conveyed subject to the terms of this Declaration, which shall run with the Properties and shall be binding upon and inure to the benefit of all Persons (defined in Section 1.1) having or acquiring any right, title, or interest in or to the Properties or any part thereof.

#### 1. Definitions.

#### 1.1 <u>Certain Defined Terms.</u>

As used in this Declaration, the terms set forth below shall have the following respective meanings:

"Area of Common Responsibility" means: (i) the Common Area, (ii) any additional areas which by the terms of this Declaration, any Supplemental Declaration, any other applicable covenants, or any contract become the responsibility of the Association, and (iii) prior to Turnover, any additional areas which Declarant from time to time designates as an "Area of Common Responsibility," the location of which additional areas may be modified, supplemented, changed, redesignated or undesignated from time to time by Declarant in its sole discretion.

Page 1 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

"Articles of Incorporation" or "Articles" means the Articles of Incorporation of the Association, as amended and in effect from time to time.

"Association" means River Bend Master Owners Association, Inc., to be formed as an Oregon nonprofit mutual benefit corporation, and its successors and assigns.

"Board of Directors" or "Board" means the board of directors of the Association, selected as provided in the Bylaws.

"Bylaws" means the bylaws of the Association, as amended and in effect from time to time.

"City" means the City of Bend, Oregon.

"Class A Members" has the meaning set forth in Section 9.3.1.

"Class B Member" has the meaning set forth in Section 9.3.2.

"Common Area" means: (i) all real and personal property that the Association owns, leases, or otherwise holds possessory or use rights in for the common use and enjoyment of the Owners, and (ii) those areas of the Properties and such personal property located thereon that are owned by Declarant and designated from time to time by Declarant as "Common Area," the location of which may be modified, supplemented, changed, redesignated, or undesignated from time to time by Declarant in its sole discretion; provided, however, no later than the date of the Turnover Meeting, Declarant shall, pursuant to Section 10.2 of this Declaration, convey to the Association any areas owned by Declarant which are then designated as Common Area.

"Common Expenses" means the actual expenses incurred and the estimated expenses anticipated to be incurred by the Association pursuant to the River Bend Documents or for the general benefit of the Properties or an Area of Common Responsibility, including any reasonable reserves, all as determined by the Board pursuant to the River Bend Documents, provided, however, prior to Turnover, Common Expenses shall not include the initial construction of Improvements on Areas of Common Responsibility or Limited Areas of Common Responsibility.

"Condemnation" and grammatical variations thereof, means a taking by any authority having the power of eminent domain or a conveyance in lieu of and under a bona fide threat of such a taking.

"Condominium Association" means the condominium owners association formed pursuant to the Oregon Condominium Act with respect to any Condominium Project.

"Condominium Documents" means the condominium declaration and other documents creating and governing any Condominium Project.

"Condominium Class A Members" means the Class A Members who are Owners of Condominium Units.

Page 2 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

"Condominium Project" means any condominium project created pursuant to the Oregon Condominium Act and developed on the Properties.

"Condominium Unit" means a condominium unit within any Condominium Project.

"County" means Deschutes County, Oregon.

"County Records" means the official records of Deschutes County, Oregon.

"CPI" means the Consumer Price Index for Urban Wage Earners and Clerical Workers (1993-95=100) U.S. City Average for All Items, as published by the United States Department of Labor, Bureau of Labor Statistics, or if such index is discontinued or revised, such other governmental index or computation with which it is replaced or, if no replacement index exists, the Association shall select as a replacement index that index which, in the Association's business judgment, is generally recognized as a successor index.

"Declarant" means River Bend Limited Partnership, an Oregon limited partnership, or any successor or assign who has or takes title to any portion of the First Phase or any subsequent Phase for the purpose of development and/or resale in the ordinary course of its business and who is designated as a Declarant in a written instrument executed by an immediately preceding Declarant and recorded in the County Records.

"<u>Design Guidelines</u>" means the design guidelines and procedures adopted by the Design Review Committee pursuant to Section 6, as amended or supplemented and in effect from time to time.

"<u>Design Review Committee</u>" or "<u>Committee</u>" means the committee appointed pursuant to Section 6.7.

"Eligible Holder" has the meaning set forth in Section 16.2.

"First Mortgage" means a Mortgage recorded in the County Records, made in good faith and for value, and having a first priority over other Mortgages, if any, on a Parcel or Condominium Unit.

"First Phase" means the real property described on Exhibit A, initially subjected to this Declaration by Declarant by execution of this Declaration and its recordation in the County Records.

"Improvement" means any structure or improvement of any kind, including buildings, fences, walls, trees, hedges, plantings, poles, driveways, parking areas, loading areas, ponds, lakes, recreational facilities, signs, changes in any exterior color or shape, and site work (including excavation, grading, road construction, utility improvements, and removal of trees or plantings); provided that "Improvement" does not include turf, shrub, or tree maintenance or replacement.

Page 3 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

"Limited Area of Common Responsibility" means: (i) Limited Common Area, (ii) any additional areas which by the terms of this Declaration, any Supplemental Declaration, any other applicable covenants, or any contract become the responsibility of any Sub-Association, and (iii) any additional areas which Declarant from time to time designates as a "Limited Area of Common Responsibility," the location of which additional areas may be modified, supplemented, changed, redesignated, or undesignated from time to time by Declarant in its sole discretion (and at the time of designation, the Declarant shall also specify the Sub-Association responsible for such "Limited Area of Common Responsibility").

"Limited Common Area" means: (i) any real or personal property that the Association or any Sub-Association owns, leases, or otherwise holds possessory or use rights in for the common use and enjoyment of the Owners of Parcels in one or more Phases, but less than all of the Properties and (ii) those areas of the Properties and such personal property located thereon that are owned by Declarant and designated from time to time by Declarant as "Limited Common Area," the location of which may from time to time be modified, supplemented, changed, redesignated, or undesignated from time to time by Declarant in its sole discretion (and at the time of designation, the Declarant shall also specify the Sub-Association responsible for such "Limited Common Area").

"Maximum Base Assessment Amount" with respect to any Member means an amount equal to Five Hundred Fifty Dollars (\$550.00) per acre of land owned by such Member (any partial acres to be rounded up to the nearest acre), which amount shall be increased by:
(i) any increase in the Maximum Base Assessment Amount agreed to under Section 11.6 below, and (ii) commencing on January 1, 2000 and on each January 1 thereafter (each, an "Adjustment Date"), two hundred percent (200%) of the percentage increase (if any) in the CPI during the calendar year immediately preceding the applicable Adjustment Date.

"Maximum Developable Acreage" means the total number of acres in the First Phase and the Potential Expansion Property less the total number of acres which are: (i) removed from the Potential Expansion Property by Declarant, (ii) removed from the coverage of this Declaration pursuant to Section 2.2, (iii) dedicated to and accepted by any public authority, or (iv) conveyed to the Association or any Sub-Association as Common Area or Limited Common Area, respectively.

"Member" means a Person entitled to membership in the Association, as provided in Section 9.2.

"Mortgage" means a mortgage, a trust deed, or a real estate contract utilized as a financing device.

"Mortgagee" means a mortgagee, a beneficiary of a trust deed, or a vendor under a real estate contract utilized as a financing device.

"Owner" means the Person(s) who holds the record title to any Parcel or Condominium Unit, excluding any Person(s) who holds an interest merely as security for the performance of an obligation. If a Parcel or Condominium Unit is sold pursuant to a real estate

Page 4 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

contract utilized as a financing device, then, upon the recording of such contract or a memorandum thereof, the vendee thereunder (rather than the fee owner) will be considered the Owner of such Parcel or Condominium Unit.

"Parcel" means any portion of the Properties, whether improved or unimproved, which may legally be independently owned and conveyed and which is intended for development, use, and occupancy for commercial, industrial, or residential purposes, including all Improvements thereon; provided that, with respect to any Condominium Project, "Parcel" means entire portion of the Properties on which the Condominium Project, including any common or limited common elements, is located and not any Condominium Unit therein; and provided further that "Parcel" does not include any of the Common Area, any Limited Common Area, or any property dedicated to and accepted by any public authority.

"Person" means a natural person, a corporation, a partnership, a limited liability company, an unincorporated association, a trust, or any other legal entity.

"Phase" means all Parcels simultaneously subjected to this Declaration by Declarant by execution and recordation of this Declaration or any Supplemental Declaration in the County Records.

"<u>Plat</u>" means any plat or planned development map affecting the Properties or any part thereof and recorded in the County Records, as such plat or map may be modified of record from time to time.

"Potential Expansion Property" means the real property described on Exhibit B, whether or not such property has been annexed to River Bend as provided herein.

"<u>Properties</u>" means the First Phase, together with any additional property annexed and thereby subjected to this Declaration in accordance with Section 2.1.

"Regular Class A Members" means the Class A Members other than the Condominium Class A Members.

"River Bend" means the multi-phase mixed-use project to be developed on the Properties.

"River Bend Documents" means this Declaration, any Supplemental Declarations, the Articles, the Bylaws, the Design Guidelines, the Use Guidelines and Restrictions, and any other rules and regulations adopted by the Association.

"67% Vote" means the affirmative vote of Class A Members together holding at least 67% of the total votes held by Class A Members.

"Special Assessment" means an assessment levied pursuant to Section 11.4.

"Specific Assessment" means an assessment levied pursuant to Section 11.5.

Page 5 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

"Sub-Association" means any separate association of Owners, as established pursuant to a Supplemental Declaration in accordance with Section 2.1.3(b).

"Supplemental Declaration" means any amendment or supplement to this Declaration recorded pursuant to Section 2.1 which annexes and thereby subjects additional property to this Declaration and any other amendment or supplement to this Declaration recorded in accordance with this Declaration.

"<u>Turnover</u>" means the turnover of control of the Association from Declarant to the Owners, which shall occur at the time and in the manner set forth in Section 9.5.

"<u>Turnover Meeting</u>" means the meeting of the Association called pursuant to Section 9.5 for the purpose of effecting the Turnover.

"Use Guidelines and Restrictions" means any rules and regulations adopted by the Board pursuant to Section 4.3 or any Supplemental Declaration, establishing requirements and restrictions regarding uses of and conduct on the Properties or any part thereof.

## 1.2 Liberal Construction.

The provisions of this Declaration shall be liberally construed to effect its purpose of creating a uniform plan for the development and operation of River Bend in a manner which complies with all applicable laws and governmental regulations and requirements.

# 1.3 Interpretation.

The captions or headings of Sections of this Declaration are intended for ease of reference only and shall have no effect whatsoever on the construction or interpretation of any provision of this Declaration. The term "including," as used in this Declaration, is not limiting and means "including without limitation." Unless otherwise specified, the words "hereof," "herein," and "hereunder" and similar words refer to this Declaration as a whole and not to any particular provision of this Declaration. References to any Section or Exhibit are references to Sections of or Exhibits to this Declaration, unless otherwise specified. All of the terms and words used in this Declaration regardless of the number used, shall be deemed and construed to include any other number (singular or plural), as the context or sense of this Declaration or any paragraph or clause hereof may require, the same as if the words had been fully and properly written in the number.

#### 1.4 Amendment and Restatement.

This Declaration amends and restates the Original Declaration in its entirety and, upon the execution and acknowledgment of this Declaration and the recordation of this Declaration in the Official Records of Deschutes County, Oregon, the Original Declaration shall be null and void and of no further force and effect.

# 2. Annexation and Withdrawal of Property.

#### 2.1 Annexation.

# 2.1.1 Annexation Without Approval of Members.

At any time and from time to time prior to Turnover, Declarant may in its sole discretion unilaterally subject all or any part of the Potential Expansion Property to the provisions of this Declaration by recording in the County Records a Supplemental Declaration containing a legal description of the property being annexed and specifically subjecting it to the terms of this Declaration. Any such Supplemental Declaration: (i) shall not require the consent of the Members or of any Mortgagees, but shall require the consent of the owner of the annexed property, if other than Declarant; and (ii) shall be signed by Declarant and by the owner of the annexed property, if other than Declarant. Any annexation pursuant to this Section 2.1.1 shall be effective upon recordation in the County Records of the Supplemental Declaration relating to such annexation, unless otherwise provided therein.

## 2.1.2 <u>Annexation With Approval of Members.</u>

At any time and from time to time, the Association or Declarant may subject any real property, including any Potential Expansion Property, to the provisions of this Declaration with: (i) the consent of the owner of such property, (ii) the approval of the Class A Members by a 67% Vote, (iii) the approval of Mortgagees holding Mortgages on Parcels the Owners of which together hold at least 67% of the total votes held by Class A Members, and (iv) if prior to Turnover, the consent of Declarant. Any such annexation shall be accomplished by recording in the County Records a Supplemental Declaration containing a legal description of the property being annexed and specifically subjecting it to the terms of this Declaration. Any such Supplemental Declaration shall be signed by the President and Secretary of the Association and by the owner of the annexed property. Any annexation pursuant to this Section 2.1.2 shall be effective upon recordation in the County Records of the Supplemental Declaration relating to such annexation, unless otherwise provided therein.

## 2.1.3 Additional Covenants and Easements.

(a) At any time and from time to time prior to Turnover, Declarant may unilaterally (but with the consent of the owner of the property in question, if other than Declarant) subject any portion of the property subjected to this Declaration initially or by Supplemental Declaration pursuant to Section 2.1.1 to additional covenants and easements, including covenants requiring the Association to maintain and insure such property; provided that the foregoing right of Declarant shall be subject to the provisions of Section 4.2 and any other applicable limitations set forth in the River Bend Documents. Any additional covenants and easements pursuant to this Section 2.1.3(a) shall be set forth in a Supplemental Declaration recorded in the County Records either concurrently with or after the annexation of the property in question.

(b) Any Supplemental Declaration pursuant to Section 2.1.1, 2.1.2, or 2.1.3(a) may provide for, without limitation: (i) the creation of Limited Common Area or any Limited Area of Common Responsibility within any one or more Phases, but less than all of the Properties; (ii) the establishment of Sub-Associations with respect to any one or more Phases, but less than all of the Properties, the purpose of which shall be to own, maintain, and insure any Limited Common Area or any Limited Area of Common Responsibility in such Phase(s); (iii) the assessment of expenses which relate solely to such Limited Common Area and Limited Area of Common Responsibility against only those Parcels located in such Phase(s); and (iv) such additional covenants as may be necessary or appropriate to implement any of the foregoing.

# 2.2 Withdrawal of Property.

Declarant reserves the right, without prior notice and without the consent of any Person, to amend this Declaration at any time prior to Turnover for the purpose of removing property then owned by Declarant, its affiliates, or the Association from the coverage of this Declaration if and to the extent that: (i) such property was originally included in error, or (ii) Declarant's plans for development of the Properties no longer include such property, so long as the withdrawal of such property is not unequivocally contrary to the overall, uniform scheme of development for the Properties. Any withdrawal of property pursuant to this Section 2.2 shall be reflected in a Supplemental Declaration signed by Declarant and recorded in the County Records.

# 2.3 Assignment of Rights.

At any time and from time to time, Declarant may transfer or assign its right to annex or withdraw property in accordance with this Section 2 to any Person who is the developer of at least a portion of the First Phase or the Potential Expansion Property. Any such transfer or assignment shall be reflected in a written instrument signed by Declarant and recorded in the County Records.

# 3. Property Rights.

#### 3.1 Common Area.

Each Owner shall have a non-exclusive right and easement of use, access, and enjoyment in and to the Common Area, which easement shall be appurtenant to and shall pass with the title to every Parcel and Condominium Unit, subject to: (i) the River Bend Documents; (ii) any restrictions or limitations contained in any deed conveying property included in the Common Area to the Association; (iii) the rights of Declarant and the Board pursuant to this Declaration to adopt rules regulating the use and enjoyment of the Common Area; (iv) the rights of Declarant and the Board to acquire or dispose of portions of the Common Area subject to any applicable limitations contained in this Declaration, (v) the right of the Association to mortgage, pledge, or hypothecate any or all of the Common Area as security for money borrowed or debts incurred, subject to the approval requirements set forth in the River Bend Documents; (vi) the right of the Association to grant temporary licenses to come upon the Properties in connection

with special events held thereon; and (vii) the right of Declarant and the Board pursuant to this Declaration to designate portions of the Common Area as Limited Common Area.

#### 3.2 Limited Common Area.

The rights and easements of use and enjoyment with respect to any Limited Common Area and any Limited Area of Common Responsibility created pursuant to Section 2.1.3(b) shall be set forth in the Supplemental Declaration creating such Limited Common Area and such Limited Area of Common Responsibility.

#### 3.3 Delegation.

Subject to the limitations set forth in Section 3.1 and any other applicable restrictions or limitations set forth in the River Bend Documents, any Owner may delegate such Owner's right and easement of use, access, and enjoyment in and to the Common Area to such Owner's family members, guests, invitees, employees, tenants, and licensees.

#### 4. Restrictions on Use.

#### 4.1 General Applicability.

Declarant intends to create River Bend as a commercial, industrial, and residential development and, in furtherance of its and every other Owner's interests, has established a general plan of development for River Bend as a master planned community. The Properties are and will be subject to architectural, design, and land development guidelines as set forth in Section 6. The Properties will also be subject to guidelines and restrictions regarding use and conduct as provided in this Section 4. All provisions of this Declaration, of any Supplemental Declaration containing any such guidelines and restrictions, and of any Use Guidelines and Restrictions shall apply to all occupants of any Parcel or Condominium Unit and to all family members, guests, invitees, employees, tenants, and licensees of any Owner, except as otherwise provided therein.

# 4.2 <u>Declarant Authority</u>.

At any time and from time to time prior to the first sale of any Parcel in the First Phase, Declarant shall have the authority, subject to the provisions of Section 4.7, to establish requirements and restrictions with respect to the use of and conduct on Parcels contained in the First Phase or any subsequent Phase. At any time and from time to time prior to the first sale of any Parcel in any Phase after the First Phase, Declarant shall have the authority, subject to the provisions of Section 4.7, to establish requirements and restrictions with respect to the use of and conduct on Parcels contained in such Phase or any subsequent Phase. Any requirements and restrictions established by Declarant pursuant to this Section 4.2 shall be set forth in a Supplemental Declaration recorded in the County Records.

# 4.3 Board Authority.

Subject to the provisions of this Section 4, the Board shall have the authority to adopt and thereafter to modify or supplement from time to time Use Guidelines and Restrictions. The Board shall send a copy of any proposed new, modified, or supplemented rule or regulation to each Owner at least thirty (30) days prior to the Board meeting at which such rule or regulation will be considered. Members shall have a reasonable opportunity to be heard at any such meeting. If any such rule or regulation is approved by the Board, it shall become effective unless disapproved at such meeting by Class A Members by a 67% Vote or by Declarant, so long as Declarant owns any property in the First Phase or any Potential Expansion Property. The Board shall provide a copy of the Use Guidelines and Restrictions then in effect to any Member or Mortgagee upon request and without cost.

#### 4.4 Members' Power.

At a meeting duly called for such purpose as provided in the Bylaws, the Class A Members may adopt, repeal, modify, limit, and expand Use Guidelines and Restrictions by a 67% Vote and with the approval of Declarant, so long as Declarant owns any property in the First Phase or any Potential Expansion Property.

# 4.5 Owners' Acknowledgment.

By accepting a deed to or entering into a recorded contract of sale for a Parcel or a Condominium Unit, each Owner acknowledges and agrees that the use, enjoyment, and marketability of such Owner's property may be affected by the provisions of this Section 4, by any requirements and restrictions established pursuant to Section 4.2, and by the Use Guidelines and Restrictions, and that any of the foregoing may be modified or supplemented from time to time as provided in this Section 4.

# 4.6 General Restrictions.

Each Parcel and Condominium Unit may be used only for purposes permitted by applicable law and in a manner complying with any Supplemental Declaration applicable thereto and adopted pursuant to Section 4.2 and with any applicable Use Guidelines and Restrictions promulgated pursuant to Section 4.3 or 4.4. Subject to the limitations set forth in this Section 4, the requirements and restrictions established pursuant to Section 4.2 and the Use Guidelines and Restrictions may address, without limitation, any or all of the following matters relating to the Properties or any portion thereof: Signage, parking, utility and landscaping installation, garbage and trash disposal, exterior antennas and similar apparatus, exterior lighting, the maintenance and repair of exteriors and landscaping, view obstruction, the conduct of business on Parcels limited to residential use, and short-term leasing activities with respect to residential properties.

#### 4.7 Limitations.

The authority of Declarant, the Board, and the Members to adopt requirements or restrictions regarding use or conduct or Use Guidelines and Restrictions shall be subject to the

Fage 10 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

following limitations: (i) similarly situated Owners and occupants shall be generally treated similarly; (ii) no rule shall interfere with the freedom of occupants of residential Parcels or Condominium Units to determine the composition of their households, except that all occupants of such a Parcel or Condominium Unit may be required to be members of a single housekeeping unit; (iii) no rule shall interfere with the activities carried on within the confines of the structures on any Parcel or Condominium Unit, except that the Association may: (a) prohibit activities not normally associated with property restricted to the use applicable to such Parcel or Condominium Unit, and (b) restrict or prohibit activities that create the possibility of monetary costs for the Association or other Owners, that endanger the health or safety of occupants of other Parcels or Condominium Units, that generate excessive noise or traffic, that create unsightly conditions visible outside the confines of such structures, or that create a source of unreasonable annoyance to occupants of other Parcels or Condominium Units; (iv) the Association shall not impose any fee for the transfer of any Parcel or Condominium Unit in an amount greater than the costs estimated to be incurred by the Association in connection with the transfer; and (v) no rule or regulation adopted or action taken by the Association shall unreasonably impede or interfere with Declarant's right to develop the Properties in accordance with authorizations obtained from the City or County or from any other applicable governmental authority.

#### 4.8 <u>Development</u>.

Nothing in this Section 4, in any Supplemental Declaration pursuant to Section 4.2, or in any of the Use Guidelines and Restrictions shall be construed to prohibit or unreasonably impede or interfere with the activities of Declarant or Declarant's employees, agents, independent contractors, successors, and assigns in connection with the development and construction of Improvements on, or the providing of utility services to, the Properties or other real property owned by Declarant, or the maintenance on portions of the Properties of such facilities as may be reasonably required or convenient in connection with such activities, including business offices, storage areas, construction yards and equipment, signs, and sales offices; provided, however, that no such activity shall be performed and no such facility shall be maintained on any portion of the Properties in such a way as to interfere unreasonably and substantially with the use, enjoyment, or access of any Owner or any Owner's family members, guests, invitees, employees, tenants, and licensees.

#### 5. Fasements.

#### 5.1 Easements for Encroachments.

Reciprocal appurtenant easements are hereby reserved for any encroachment, and for maintenance and use of any encroachment permitted under this Section 5.1, which now or hereafter exists between any Parcel and any adjacent Common Area or between any adjacent Parcels, to the extent that any such encroachment: (i) is not the result of any willful, knowing, or negligent conduct on the part of an Owner or occupant or the Association, as the case may be; and (ii) extends for a distance of not more than three feet, measured from any point on the common boundary along a line perpendicular to such boundary.

#### 5.2 Easements for Utilities, Etc.

Access and maintenance easements are hereby reserved for 5.2.1 Declarant (so long as Declarant owns any property in the First Phase or any Potential Expansion Property), the Association, and the designees of each (which may include the City, the County and any utility company) on, across, over, and under all of the Properties to the extent reasonably necessary for the purpose of installing, replacing, repairing, and maintaining roads, walkways, bicycle pathways, trails, ponds, wetlands, drainage systems, street lights, signage, cable television systems, master television antenna systems, security and similar systems, and all other utilities (including water, sewer, telephone, gas, and electricity) and facilities and equipment related to any of the foregoing; provided that all of the foregoing shall be installed only on: (i) "Private Common Ways" (as such term is used on applicable Plats), (ii) Common Areas, (iii) Limited Common Areas, (iv) property owned by the installing party, or (v) within easements designated for such purposes on the applicable Plat. The easements reserved pursuant to this Section 5.2.1 shall not entitle the holders thereof to install any of the foregoing systems or facilities over, under, or through any structure existing on a Parcel at the time of installation, and any damage to any Improvement resulting from the exercise of any such easement shall promptly be repaired by, and at the expense of, the Person exercising the easement. The exercise of any easement reserved pursuant to this Section 5.2.1 shall not interfere unreasonably and substantially with the use of any Parcel or Condominium Unit and, except in an emergency, entry onto any Parcel or into any Condominium Unit shall be made only after reasonable notice to the Owner or occupant thereof. To the extent practicable, all such easements affecting a Parcel shall be sited within the setback areas designated for such Parcel.

Declarant hereby specifically grants to any water supplier, 5.2.2 electric company, telephone company, sewer utility, cable television provider, and natural gas supplier serving the Properties easements across the Properties for the purposes of installing, replacing, repairing, maintaining underground utility lines at least twelve (12) inches beneath the surface, and reading utility meters and boxes; provided that the exercise of such easement shall not permit entry into any structure on a Parcel, nor shall any utilities be installed or relocated on the Properties, without the prior approval of the Declarant (prior to Turnover) or Board and, if any utility is proposed to be installed or relocated on a Parcel or Condominium Unit, the approval of the Owner of the affected Parcel or Condominium Unit; provided, further, in no event (other than an emergency) may any utility be installed, relocated or repaired on the portions of the Properties used for retail sales during the period commencing on September 15 and ending on December 31. To the extent practicable, all such easements affecting a Parcel shall be sited within the setback areas designated for such Parcel. In the event that any grantee granted an easement pursuant to this Section 5.2.2 requests a specific easement by separate recordable document, Declarant (prior to Turnover) and the Association are each hereby granted the right and authority to grant such easement in accordance with the provisions of this Section 5.2.2. The grantor of a utility easement retains the right to require the relocation of any such easements at the grantee's sole cost and expense, to use the surface of the easement property and to allow other utility lines to be installed within the easement property, provided that the utility lines do not materially interfere with the installations of the grantee. In its use of the easement

property, the grantee shall replace and restore, at the grantee's sole cost and expense, any portions of the Properties disturbed by such work.

# 5.3 Easements for Construction and Sales Activities.

Declarant, for itself and its agents, representatives, employees, successors, assigns, licensees, and Mortgagees, hereby reserves an easement over, in, on, under, and across the Common Area to make such other use thereof (including the storage of equipment and materials) as may be reasonably necessary or incidental to the development of and construction of Improvements on the Properties or on other real property owned by Declarant (including the Potential Expansion Property, whether or not such property is annexed and thereby made subject to this Declaration) or to the sale and marketing of Parcels and Improvements thereon (including Condominium Units); provided, however, that no such activity shall be performed in such a way as to interfere unreasonably and substantially with the use, enjoyment, or access of any Owner or any Owner's family members, guests, invitees, employees, tenants, and licensees.

# 5.4 Easements for Drainage.

Easements are hereby reserved over every Parcel and the Common Area for natural drainage of storm water runoff from other portions of the Properties and other properties adjacent thereto owned by Declarant or its successors. In addition, an easement is hereby reserved for Declarant (so long as Declarant owns any property in the First Phase or any Potential Expansion Property), the Association, and the designees of each to enter on, across, over, in, and under any portion of the Properties for the purpose of changing, correcting, or otherwise modifying the grade or drainage channels of the Properties for the purpose of improving drainage from and across the Properties; provided that the holders of such easement shall use all reasonable efforts to conduct any such work in a manner which minimizes any disturbance to the uses of the Properties by Declarant, the Association, the Owners, and the Owners' respective family members, guests, invitees, employees, tenants, and licensees; shall undertake any such work expeditiously; and shall restore any areas affected by such work to a sightly and usable condition as soon as reasonably possible following the completion such work. No Person shall alter the natural drainage on or over any Parcel so as to increase materially the drainage of storm water onto adjacent portions of the Properties without the consent of the Owner of the affected property.

# 5.5 Right of Entry.

The Association shall have the right, but not the obligation, to enter upon any Parcel, including any Improvements thereon, for emergency, security, or safety reasons (including the correction of any condition which may increase the possibility of a fire or other hazard in the event an Owner fails or refuses to cure the condition within a reasonable time after request by the Board), to perform maintenance pursuant to this Declaration or any Supplemental Declaration, and to determine whether such Parcel and Improvements and the activities thereon comply with the River Bend Documents. Such right may be exercised by the Board, the Design Review Committee, officers, agents, employees, and/or managers of the Association, and all policemen, firemen, ambulance personnel, and similar emergency personnel in the performance

Page 13 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

of their duties. Except in an emergency situation, entry to any structure shall occur only during reasonable hours and after reasonable advance notice to the Owner thereof.

# 5.6 Easements for Vehicular Ingress, Egress and Parking.

Easements are hereby reserved over every Parcel, Common Area, and Limited Common Area for: (i) vehicular ingress and egress to, from, over and across any roads, streets or driveways located on any portion of the Properties, and (ii) vehicular parking in any parking lot striped for parking that may from time to time be located on the Properties, subject to: (i) any restrictions contained in the Use Guidelines and Restrictions, (ii) the right of any Owner, at its sole cost and expense, to permanently designate and mark 25% of its parking spaces to be used during its usual business hours exclusively for its own purposes (such Owner shall be entitled, at its sole cost and expense, to enforce restrictions respecting unauthorized parking in such parking spaces), and (iii) the obligation of each Owner to instruct its employees to park in its parking spaces (which obligation shall be in addition to and not in lieu of any Use Guidelines and Restrictions). Such easements are for the benefit of Declarant and its partners, tenants, franchisees, subtenants, concessionaires, successors and assigns, the Owners, and each of their respective agents, licensees, consultants, suppliers, family members, patrons, customers, guests, employees and invitees.

# 5.7 Easements Deemed Created.

Each conveyance of a Parcel or a Condominium Unit after the date of recordation of this Declaration shall be construed to grant and reserve the easements contained in this Section 5 and elsewhere in the River Bend Documents, regardless of whether any specific reference to such easements appears in the instrument of conveyance.

# 6. Design Review Committee.

# 6.1 <u>Design Review</u>.

No Improvement shall be commenced, erected, placed, altered, added to, or maintained on, within, beneath, or above the Properties or any portion thereof until and unless design plans and specifications showing the site layout, structural design, exterior elevations, exterior materials and colors, signs, landscaping, drainage, lighting, irrigation, utility facilities layout, and screening for such Improvements have been submitted to and approved in writing by the Design Review Committee; provided that: (i) any Owner may remodel, paint, or redecorate the interior of structures on such Owner's Parcel or the interior of such Owner's Condominium Unit without such approval except to the extent, if any, that such modifications are visible from outside the structure; and (ii) no such approval shall be required to repaint the exterior of a structure in accordance with the originally approved color scheme or to rebuild any Improvement in accordance with originally approved plans and specifications. The intent and purpose of this Declaration is to achieve a high standard of quality of workmanship and materials and to assure harmony of external design with existing Improvements and location with respect to topography and finished grade elevations.

#### 6.2 Design Guidelines.

The Design Review Committee may at any time and from time to time adopt and issue, and thereafter supplement and modify, Design Guidelines which more specifically define and describe the design standards for River Bend and the various land uses therein and which set forth the procedure and specific requirements for Committee approval of proposed Improvements pursuant to this Section 6. The Design Guidelines may vary for different Phases. The Design Guidelines shall not in any event be inconsistent with this Declaration.

#### 6.3 Governmental Approvals.

Compliance with the design review process set forth in this Declaration is not a substitute for compliance with City or County building, zoning, and subdivision regulations, and each Owner shall be responsible for obtaining all approvals, licenses, and permits required by applicable law prior to commencing construction. Approval of any Improvement by the Committee does not assure approval by the City or County or any other governmental agency.

#### 6.4 Variance.

The Committee may authorize variances from compliance with any of its Design Guidelines when circumstances such as topography, natural obstructions, hardship, or aesthetic or environmental considerations require; provided that such variances may be granted only when unusual circumstances dictate; and provided further that no variance shall: (i) be effective unless in writing; (ii) be contrary to this Declaration; or (iii) limit the Committee's authority to deny a variance in other circumstances. For purposes of this Section 6.4, the inability of an Owner to obtain any governmental approval, the issuance of any permit, the cost of compliance, or the terms of any financing shall not be considered a hardship warranting a variance.

#### 6.5 Committee Decision.

The Committee shall render its decision on an application for approval pursuant to this Section 6 within thirty (30) days after it has received the complete written application. Each application shall specify the approval requested and be accompanied by all material reasonably requested by the Committee to enable it to make an informed decision on the application. The plans and specifications included with each application shall be prepared by a licensed architect. The Committee may charge a reasonable fee to cover the estimated cost of processing an application for any approval pursuant to this Section 6.

#### 6.6 Committee Discretion.

The Committee may, at its sole discretion, withhold consent to any proposed Improvement if the Committee finds the proposed Improvement would be inappropriate for the particular Parcel or incompatible with the design standards that the Committee establishes for River Bend or for the Phase in which the Parcel in question is located. The Committee may take into account any factors its reasonably deems relevant with respect to any proposed Improvement, including siting, shape, size, color, design, height, solar access, impairment of the

view from other Parcels, effect on the enjoyment of other Parcels or the Common Area, disturbance of existing terrain and vegetation, and environmental impact.

# 6.7 Membership: Appointment and Removal.

The Committee shall consist of as many persons, but not fewer than three, as Declarant (prior to Turnover) or the Board (after Turnover) may appoint from time to time. Declarant (prior to Turnover) or the Board (after Turnover) may remove any member of the Committee at any time, with or without cause, and may appoint new or additional members at any time. The Association shall keep on file at its principal office a list of the names and addresses of the members of the Committee. If Declarant or the Board, as applicable, fails to appoint members of the Committee, the Board shall serve as the Committee.

# 6.8 Majority Action.

The affirmative vote of a majority of the members of the Committee shall govern its actions and constitute the act of the Committee. A quorum of the Committee shall consist of a majority of the Committee's members. The Committee may render its decision only by written instrument setting forth the action taken by the members.

# 6.9 Limitation of Liability.

The Committee shall use reasonable judgment in approving or disapproving applications submitted to it. Neither Declarant, the Association, the Board, the Committee, nor any member, manager, employee, agent, or consultant of any of the foregoing shall be: (i) liable to any Person, including any Owner, occupant, or contractor: (a) for any official act of the Committee in connection with a submitted application, except to the extent the Committee or any individual Committee member acts with malice or wrongful intent; or (b) for any loss, liability, claim, or expense which may arise by reason of the approval or disapproval of any Improvement; or (ii) responsible in any way for any defects in any plans or specifications submitted, revised, or approved pursuant to this Section 6 or for any structural or other defect in any work done. The Association shall indemnify, hold harmless, and defend the Committee and its members for, from and against any and all claims, losses, liabilities, and expenses (including reasonable attorneys' fees) incurred in connection with any suit, action, or proceeding which results from any action of the Committee pursuant to this Section 6; provided that the Association shall not be obligated to indemnify any member of the Committee to the extent such member is adjudged to be liable for misconduct in the performance of such member's duties, unless (and then only to the extent that) the tribunal in which such suit or proceeding is brought determines that, despite the adjudication of liability but in view of all circumstances of the case, such member is fairly and reasonably entitled to indemnification. The Association shall use all reasonable efforts to procure errors and omissions insurance coverage for members of the Committee.

#### 6.10 Nonwaiver.

Approval by the Committee of any application shall not be deemed to constitute a precedent or waiver impairing its right to withhold approval as to any similar application thereafter submitted.

#### 6.11 Effective Period of Consent.

The Committee's approval of any proposed Improvement shall automatically be revoked one year after issuance unless construction of the Improvement has been commenced or the Owner has applied for and received an extension of time from the Committee.

#### 6.12 Estoppel Certificate.

Within ten (10) business days after written request therefor is delivered to the Committee by any Owner, and upon payment to the Committee of a reasonable fee, if any, fixed by the Committee to cover its estimated costs, the Committee shall provide such Owner with an estoppel certificate executed and acknowledged by a member of the Committee, certifying with respect to any Parcel or Condominium Unit owned by the Owner whether or not, as of the date thereof, the Improvements on or comprising such Parcel or Condominium Unit comply with this Declaration. If the estoppel certificate states that any Improvements do not so comply, such certificate shall also identify the noncomplying Improvements and set forth the nature of such noncompliance in reasonable detail. Any purchaser from the Owner and any Mortgagee of the Parcel or Condominium Unit in question shall be entitled to rely conclusively on such certificate with respect to the matters set forth therein.

## 6.13 Activities of Declarant.

The provisions of this Section 6 shall not apply to the activities of Declarant or its affiliates, or to Improvements to the Common Area undertaken by or on behalf of the Association.

7. Construction and Alteration of Improvements; Sewer Services and Irrigation Water.

#### 7.1 General.

The Design Guidelines and the general restrictions set forth in this Declaration shall govern the construction, reconstruction, refinishing, alteration, and maintenance of any Improvement upon, under, or above the Properties or any part thereof.

#### 7.2 Construction Methods.

All Owners shall comply with specific rules regarding construction methods, including excavation, drainage, utility lines, loading areas, waste storage, trash removal, materials storage, and transformers and meters, set forth in the Design Guidelines.

Page 17 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

#### 7.3 Completion of Improvements.

The construction of any building, including an Owner's dwelling (but excluding any building constructed by Declarant), including painting and all exterior finish, shall be pursued diligently and shall be completed within eighteen (18) months after the beginning of construction so as to present a finished appearance when viewed from any angle. All landscaping on a Parcel shall be completed within twelve (12) months after the date of completion of the structures thereon. In the event of undue hardship due to weather conditions, the foregoing periods may be extended for a reasonable length of time upon written approval of the Committee.

#### 7.4 Setback, Maximum Height, and Minimum Yard Requirements.

Each Parcel shall be subject to: (i) the setback, maximum height, and minimum yard requirements, if any, which are shown on the Plat depicting such Parcel or established by the governmental authority with jurisdiction over such Parcel, and (ii) such more restrictive setback, maximum height, or minimum yard requirements, if any, as may set forth in the Design Guidelines or otherwise established from time to time by the Committee. No Improvement shall be constructed or maintained in violation of any setback, maximum height, or minimum yard requirement, except with the written approval of the Committee and any applicable governmental approval.

#### 7.5 Sewer Services and Irrigation Water.

In connection with the development of the Properties, Declarant has installed or will be installing a sewer system and a system for irrigation water. To the extent a Parcel is served by such sewer system, the Owner of such Parcel shall be obligated to purchase sewer services from Declarant so long as the cost of such sewer services does not substantially exceed the amounts charged by the City for similar sewer services. To the extent a Parcel is served by such irrigation water system, the Owner of such Parcel shall be obligated to purchase irrigation water from Declarant so long as the cost of such irrigation water is not substantially more than the amounts charged by the City for potable water.

#### 8. Restrictions on Partition and Combination.

#### 8.1 Parcels.

No part of a Parcel (including a Parcel created by the combination of two or more Parcels in accordance with this Section 8.1) shall be partitioned or separated from any other part thereof, and no Parcels shall be combined, except with the prior written approval of Declarant (prior to Turnover) or the Committee (after Turnover) and full compliance with all applicable legal requirements and any applicable Condominium Documents. The approval of Declarant or the Committee may be conditioned upon payment by the Owner(s) concerned of all reasonable expenses incurred in connection with such approval, including attorneys' fees. Any agreement and recorded instrument providing for a partition or combination of Parcels shall make adequate

adjustment of voting rights and liability for assessments appurtenant to or imposed on such Parcels.

#### 8.2 Common Area.

No Owner shall bring any action for partition or division of the Common Area or any Limited Common Area. By accepting a deed to or entering into a recorded contract of sale for a Parcel or Condominium Unit, each Owner shall be deemed to have specifically waived such Owner's rights to institute or maintain any partition or other action designed to cause a division of the Common Area or any Limited Common Area. Any Owner who institutes or maintains any such action shall be liable to the Association and any affected Sub-Association for its costs and expenses in connection therewith, including their reasonable attorneys' fees.

# 9. Association Function, Membership, and Voting Rights.

#### 9.1 Function of Association.

The Association shall be responsible for: (i) the management, maintenance, operation, and control of the Area of Common Responsibility; (ii) the enforcement of this Declaration, any Supplemental Declaration, and such rules and regulations governing the use of the Properties as the Board may adopt in accordance with this Declaration or any Supplemental Declaration, including the Use Guidelines and Restrictions; and (iii) the administration and enforcement of the design standards and controls set forth in this Declaration, in any Supplemental Declaration, and in the Design Guidelines.

#### 9.2 Membership.

Every Owner shall be a Member of the Association. There shall be only one membership per Parcel or Condominium Unit. If a Parcel or a Condominium Unit is owned by more than one Person, all co-Owners shall: (i) share the privileges of the membership associated with such Parcel or Condominium Unit, subject to reasonable Board regulation and the restrictions on voting set forth in Section 9.4 and in the Bylaws, and (ii) be jointly and severally liable for the performance of the responsibilities of Owners, including the payment of assessments. The membership rights and privileges of an Owner who is a natural person may be exercised by the Member or the Member's spouse. The membership rights of an Owner which is a corporation, partnership, limited liability company, or other legal entity may be exercised by any officer, director, partner, member, manager, or trustee, or other individual designated from time to time by the Owner in a written instrument delivered to the Association. Membership in the Association shall commence, exist, and continue simply by virtue of ownership of a Parcel or a Condominium Unit, shall expire automatically upon the termination of such ownership, and need not be confirmed or evidenced by any certificate or acceptance of membership.

#### 9.3 Voting.

The Association shall have Class A Members and one or more Class B Members.

Page 19 - AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

#### 9.3.1 Class A.

The "Class A Members" shall consist of all Owners other than: (i) the Class B Member(s), if any (provided, however, once a Parcel owned by Declarant has been developed and the occupant of such Parcel is conducting business on such Parcel, Declarant shall be deemed to be a Class A Member with respect to such Parcel), and (ii) any Condominium Association. Each Regular Class A Member shall have one equal vote for each 1,000 square feet of surface area included in the Parcel owned by such Member. The Condominium Class A Members in any Condominium Project shall collectively have one equal vote for each 1,000 square feet of surface area included in the Parcel on which such Condominium Project is located. All votes held by Condominium Class A Members shall be cast only by the Condominium Association of such Condominium Project and individual Condominium Class A Members shall not be entitled to separate voting rights.

#### 9.3.2 Class B.

The "Class B Member" shall consist of Declarant. The Class B Member shall be entitled to voting rights equal to the product of: (i) the number of votes pertaining to each Parcel owned by Declarant, determined as provided in Section 9.3.1 as if Declarant were a Regular Class A Member, multiplied by (ii) three. If Declarant owns a majority of the Condominium Units in any Condominium Project, Declarant shall be deemed to own all of the Condominium Units for purposes of the immediately preceding sentence. Notwithstanding anything to the contrary contained in this Section 9.3.2, as set forth in Section 9.3.1 above, a portion of the Class B membership shall be converted to Class A membership as Parcels owned by Declarant are developed and business is conducted on such Parcels, and any remaining Class B membership shall terminate and be converted to Class A membership upon the earliest to occur of: (a) the first date on which 95% of the Maximum Developable Acreage is owned by Owners other than Declarant; (b) January 1, 2022; or (c) the date on which Declarant, in its sole discretion, terminates the Class B membership by recordation in the County Records of an instrument expressly providing for such termination.

# 9.4 Exercise of Voting Rights.

If more than one Regular Class A Member owns any Parcel, the votes attributable to such Parcel shall be exercised in such manner as the co-Owners may determine among themselves and advise the Secretary of the Association in writing prior to any meeting. In the event the Secretary is not advised to the contrary, any Owner of such a Parcel shall be deemed to have the power to cast all votes allocable to the Parcel and the Parcel's votes shall be suspended if more than one Person seeks to cast them.

#### 9.5 Turnover.

Within 120 days after termination of the Class B membership pursuant to Section 9.3.2, Declarant shall call a meeting in the manner provided in the Bylaws for the purpose of turning over administrative responsibility for the Properties to the Association. If Declarant does not call the Turnover Meeting within the required period, any two or more Members may do so

Page 20 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

by giving notice in the manner provided in the Bylaws. At the Turnover Meeting: (i) the members of the Board of Directors then holding office shall resign and their successors shall be elected as provided in the Bylaws; and (ii) Declarant shall deliver to the Association the originals of all books, records, plans, contracts, and other appropriate documents and materials in Declarant's possession relating to the Association.

# 10. Rights and Obligations of the Association.

# 10.1 Area of Common Responsibility.

Subject to the rights of Declarant and of the Owners set forth in this Declaration and in any Supplemental Declaration, the Association shall manage and control the Area of Common Responsibility, all Improvements thereon, and all fixtures and personal property related thereto, and shall keep it in good, clean, attractive, and sanitary condition, order, and repair.

# 10.2 Real and Personal Property for Common Use.

Subject to any limitations set forth in the River Bend Documents, the Association may acquire (from Declarant or any other Person), hold, mortgage or otherwise assign or pledge, and dispose of tangible and intangible real and personal property for any purpose of the Association. Declarant shall convey the Common Area to the Association no later than the date of the Turnover Meeting.

# 10.3 Enforcement.

The Association may exercise all rights and remedies set forth in the River Bend Documents in connection with the enforcement thereof. The Association may, by contract or other agreement, enforce applicable County and City ordinances and permit governmental authorities to enforce applicable ordinances on the Properties for the benefit of the Association and its Members.

# 10.4 Implied Rights: Board Authority.

The Association may exercise any other right or privilege given to it expressly by this Declaration or the other River Bend Documents, or reasonably implied from or reasonably necessary to effect any such right or privilege. Except as otherwise expressly provided in the River Bend Documents or by applicable law, all rights and powers of the Association may be exercised by the Board without a vote of the Members.

# 10.5 Property for Public Use.

So long as Declarant owns any property in the First Phase or any Potential Expansion Property, Declarant may designate sites within the Properties owned by Declarant or the Association, including portions of the Common Area, for fire, police, and utility facilities, public schools and parks, and other public facilities.

# 10.6 Professional Management; Contracts.

The Association may also enter into such contracts for goods or services as the Association deems appropriate, including, without limitation, any contracts with Declarant or any person or entity affiliated with Declarant (and if the Association enters into any such contract with Declarant or any person or entity affiliated with Declarant, the terms and provisions of any such contract shall be presumed to be fair to the Association absent manifest evidence that the terms and provisions of such contract were both: (i) commercially unreasonable at the time such contract was created, and (ii) materially more onerous to the Association at the time such contract was entered into than the terms and provisions the Association would have obtained had the Association contracted with a third-party in an arm's-length transaction).

#### 10.7 <u>Liability and Indemnification</u>.

The officers and directors of the Association, and members of committees established by the Board, shall not be liable for: (i) any mistake of judgment, negligent or otherwise, except for their own individual willful misfeasance, malfeasance, misconduct, or bad faith; or (ii) any contract or other commitment made or action taken in good faith on behalf of the Association. The Association shall indemnify, hold harmless, and defend each such officer, director, and committee member for, from and against any and all claims, losses, liabilities, and expenses (including reasonable attorneys' fees) incurred in connection with any suit, action, or proceeding which results from any action taken by or on behalf of the Association, the Board, or such committee pursuant to the River Bend Documents. Any right to indemnification provided for in this Section 10.7 shall not be exclusive of any other rights to which any present or former officer, director, or committee member may be entitled under applicable law. The Association shall use reasonable efforts to obtain errors and omissions insurance coverage for officers, directors, and committee members of the Association. In accordance with ORS 65.784, if the Association indemnifies or advances expenses pursuant to this Section 10.7 or ORS 65.391-65.401, the Association shall report the indemnification or advance in writing to the Members with or before the notice of the next meeting of the Members.

#### 10.8 Security.

The Association may, but shall not be obligated to, maintain or support certain activities within the Properties designed to make the Properties safer than they otherwise might be. NEITHER THE ASSOCIATION, THE BOARD, NOR DECLARANT SHALL IN ANY WAY BE CONSIDERED AN INSURER OR GUARANTOR OF SECURITY FOR THE PROPERTIES, NOR SHALL ANY OF THEM BE LIABLE FOR ANY LOSS OR DAMAGE RESULTING FROM ANY FAILURE TO PROVIDE ADEQUATE SECURITY OR THE INEFFECTIVENESS OF ANY SECURITY MEASURES UNDERTAKEN. NO REPRESENTATION OR WARRANTY IS MADE THAT ANY FIRE PROTECTION, BURGLAR ALARM, OR OTHER SECURITY SYSTEM CANNOT BE COMPROMISED OR CIRCUMVENTED, NOR THAT ANY SUCH SYSTEM OR OTHER SECURITY MEASURES UNDERTAKEN WILL IN ALL CASES PREVENT LOSS OR PROVIDE THE DETECTION OR PROTECTION FOR WHICH SUCH SYSTEM OR MEASURE IS DESIGNED OR

Page 22 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

INTENDED. EACH OWNER ACKNOWLEDGES, UNDERSTANDS, AND AGREES TO INFORM ITS TENANTS THAT THE ASSOCIATION, THE BOARD, AND DECLARANT ARE NOT INSURERS AND THAT EACH PERSON USING THE PROPERTIES ASSUMES ALL RISK FOR LOSS OR DAMAGE TO PERSONS AND PROPERTY RESULTING FROM ACTS OF THIRD PARTIES.

#### 11. Assessments.

#### 11.1 Creation of Assessments.

- each Parcel and Condominium Unit for Association expenses as the Board may specifically authorize from time to time, subject to the provisions of this Section 11. There shall be three types of assessments: (i) Base Assessments to fund Common Expenses; (ii) Special Assessments, as described in Section 11.4; and (iii) Specific Assessments, as described in Section 11.5. Each Owner, by accepting a deed to or entering into a recorded contract of sale for any Parcel or Condominium Unit, is deemed to covenant and agree to pay all assessments levied in accordance with this Section 11.
- together with interest, late charges, costs, and reasonable attorneys' fees as provided in this Section 11, shall be: (i) a charge and continuing lien upon such Parcel or Condominium Unit until paid, as more particularly provided in Section 11.8; and (ii) the personal obligation of the Owner(s) of such Parcel or Condominium Unit on the date the assessment is levied by the Association. Upon a transfer of title to a Parcel or Condominium Unit, the grantee shall not be personally liable for any assessments and other charges levied prior to the date of conveyance unless such liability is expressly assumed by such grantee, but such Parcel or Condominium Unit shall remain subject to the lien rights of the Association and the liability of the Owner(s) of the Parcel or Condominium Unit on the date the assessment was levied shall not be extinguished by such transfer. No holder of a First Mortgage who acquires title to a Parcel or Condominium Unit by exercising the remedies provided in its Mortgage shall be liable for unpaid assessments levied prior to the date of such acquisition of title.
- 11.1.3 Assessments shall be paid in such manner and on such dates as the Board may establish from time to time. If the Board so determines, assessments may be payable in two or more installments. Unless the Board otherwise determines, the Base Assessment shall be due and payable quarterly in advance on the first day of each fiscal quarter of the Association.
- delivered to the Association by any Owner who is liable for any assessment hereunder, the Association shall furnish to such Owner a written certificate signed by an officer of the Association stating whether such assessment has been paid. Such certificate shall be conclusive evidence of payment. The Association may require the advance payment of a reasonable processing fee for the issuance of any such certificate.

assessments, whether by nonuse of the Common Area, abandonment of such Owner's Parcel or Condominium Unit, or any other means. The obligation to pay assessments is a separate and independent covenant on the part of each Owner. No diminution or abatement of assessments or set-off shall be claimed or allowed for any alleged failure of Declarant, the Association, the Board, or the Committee to take any action or perform any function, for inconvenience or discomfort arising from the construction, alteration, repair, replacement, or maintenance of Improvements, or as a result of any other action taken by Declarant, the Association, the Board, the Committee, or their respective partners, members, managers, employees, agents, or independent contractors.

# 11.2 Determination of Base Assessment.

- level which is reasonably expected to produce total income for the Association equal to the total budgeted Common Expenses, including reserves, as determined by the Board for each fiscal year of the Association, taking into account: (i) other sources of funds available to the Association, if any; (ii) the number of Parcels and Condominium Units which are subject to assessment under Section 11.7 on the first day of the fiscal year for which the assessment is being levied; and (iii) the number of Parcels and Condominium Units reasonably anticipated to become subject to assessment during the fiscal year.
- 11.2.2 Subject to the provisions of Section 11.2.3, the Base Assessment for each year shall be allocated among all Parcels in proportion to the number of votes attributable to each Parcel pursuant to Section 9.3.1. The Base Assessment so allocated to each Condominium Project shall be allocated equally among the Condominium Units in such Condominium Project, unless the applicable Condominium Documents otherwise provide.
- 11.2.3 Subject to the provisions of Section 11.2.4, Parcels owned by Declarant or Declarant's affiliates (excluding affiliates who are natural persons) shall not be subject to payment of the Base Assessment so long as: (i) Declarant is not a Class A Member with respect to such Parcels (as determined pursuant to Section 9.3.1 above), (ii) any Parcel owned by Declarant's affiliates has not been developed and/or the occupant of such Parcel is not conducting business on such Parcel, or (iii) termination of Class B membership pursuant to Section 9.3.2 has not occurred; provided, however, that in the event that the aggregate Base Assessments levied with respect to any fiscal year ending prior to the date of the Turnover Meeting, even if paid in full by all Owners liable for such assessments, would not be sufficient to pay all actual operating expenses for which line items are included in the budget approved by the Board for that year, Declarant shall have the right (but not the obligation) to loan to the Association the amount of such deficiency, interest free. Any amount loaned to the Association by Declarant pursuant to this Section 11.2.3 shall be repaid by the Association to Declarant as soon as funds are available and to the extent the Association has available funds. If any such loan has not been repaid in full on the date of the Turnover Meeting, such loan shall be deemed forgiven and Declarant shall have no right to the repayment of such loan and the Association shall have no further obligation with respect to such loan.

11.2.4 From and after the first date on which a Condominium Unit in any Condominium Project developed by Declarant is sold to a Person other than Declarant, the Condominium Units in such Condominium Project which are owned by Declarant shall be subject to the payment of the Base Assessment in the same manner as Condominium Units owned by Condominium Class A Members, subject to any contrary provisions in the applicable Condominium Documents.

11.2.5 The Board shall send a copy of the budget for each fiscal year of the Association and notice of the amount of the Base Assessment for such year to each Owner within thirty (30) days after adoption of such budget.

#### 11.3 Reserves.

The budget for each fiscal year of the Association may include provision for reserves which take into account the number and nature of replaceable assets of the Association, the expected life of each such asset, and its expected repair or replacement cost.

#### 11.4 Special Assessments.

Upon obtaining the affirmative vote of the Class A Members holding 67% of the total votes held by such Members represented at a meeting of the Association at which a quorum is present, the Association shall have the power to levy Special Assessments from time to time to cover unbudgeted expenses or expenses in excess of those budgeted, provided however, such a vote of the Class A Members shall not be required and the Board may levy Special Assessments in either or both of the following events: (i) unbudgeted funds are required to be expended by an order of a court or governmental agency, or (ii) unbudgeted funds are necessary to repair or maintain any portion of the Area of Common Responsibility in order to prevent or reduce a threat to personal safety. For purposes of this Section 11.4, a quorum means the Class A Members holding a majority of the total votes held by all the Class A Members. Special Assessments shall be payable in such manner and at such times as may be determined by the Board, and may be payable in installments extending beyond the fiscal year to which the Special Assessment relates. Special Assessments shall be allocated in the manner provided for Base Assessments pursuant to Section 11.2.

## 11.5 Specific Assessments.

The Association shall have the power to levy Specific Assessments from time to time against any particular Parcel or Condominium Unit in order to pay: (i) costs, including overhead and administrative costs, of providing benefits, items, or services to such Parcel or Condominium Unit or occupants thereof upon request of the Owner pursuant to a menu of special services which the Board may from time to time authorize to be offered to Owners (which services may include snow removal, landscape maintenance, provision of firewood, handyman service, pool cleaning, pest control, etc.), which assessments may be levied in advance of the provision of the requested benefit, item, or service as a deposit against charges incurred by the Owner; or (ii) costs incurred in bringing the Parcel or Condominium Unit into compliance with the terms of the River Bend Documents or as a consequence of the conduct of the Owner or

occupants of the Parcel or Condominium Unit or their respective family members, guests, invitees, employees, tenants, or licensees.

# 11.6 <u>Limitation on Increases in Base Assessments.</u>

The Board may not impose Base Assessments on any Member for any fiscal year in amounts which are more than the Maximum Base Assessment Amount without obtaining the affirmative vote of the Class A Members holding 67% of the total votes held by such Members represented at a meeting of the Association at which a quorum is present. For purposes of this Section 11.6, a quorum means the Class A Members holding a majority of the total votes held by all the Class A Members.

## 11.7 Date of Commencement of Assessments.

The obligation to pay the assessments in accordance with this Section 11 shall commence as to all Parcels within each Phase on the earlier of: (i) the first day of the calendar month immediately following the first conveyance of a Parcel or Condominium Unit in such Phase to an Owner other than Declarant; or (ii) the first day of the calendar month immediately following conveyance of the Common Area in such Phase to the Association or the Limited Common Area in such Phase, if any, to the Association or the Sub-Association for such Phase, if any. The Base Assessment for the fiscal year during which the obligation to pay assessments commences as to any Phase shall be prorated for Parcels and Condominium Units in that Phase based upon the number of days remaining in such year after the date on which such obligation commences.

#### 11.8 Lien for Assessments.

11.8.1 Each assessment levied against a Parcel or Condominium Unit in accordance with this Section 11 shall constitute a lien on such Parcel or Condominium Unit. Such lien shall also secure payment of interest and late charges as provided in Section 11.13, costs of collection (including reasonable attorneys' fees) and recording fees incurred by the Association in connection with such assessment and lien, and a fee for preparing the notice of lien in such amount as the Board may determine from time to time. Such lien shall be superior to all other liens, except: (i) the liens of those taxes, bonds, assessments, and other levies which are superior by operation of law; and (ii) the lien of any First Mortgage on such Parcel or Condominium Unit. The Association may enforce such lien, when delinquent, by suit, judgment, and foreclosure.

Parcel or Condominium Unit by recording in the County Records a written statement setting forth the name of the Owner(s), the legal description of the Parcel or Condominium Unit, the name of the Association, and the delinquent assessments and other amounts then owing. Such statement shall be executed and acknowledged by an officer or property manager of the Association, and shall be served upon the Owner of the Parcel or Condominium Unit by mail in the manner provided herein for notices. The Association may proceed to foreclose the lien in accordance with applicable law at any time after the expiration of the 30-day period following

Page 26 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

the mailing of such notice. The Association may sue for unpaid assessments and costs without foreclosing or waiving the lien securing the same.

any foreclosure sale pursuant to this Section 11.8 and acquire and thereafter hold, lease, mortgage, and convey such Parcel or Condominium Unit. While a Parcel or Condominium Unit is owned by the Association following foreclosure: (i) no right to vote shall be exercised on its behalf; (ii) no assessment shall be levied on it; and (iii) each other Parcel and Condominium Unit shall be charged, in addition to its usual assessment, its pro rata share of the assessment that would have been charged such Parcel or Condominium Unit had it not been acquired by the Association.

affect any assessment lien thereon; provided, however, that the sale of any Parcel or Condominium Unit pursuant to foreclosure of a First Mortgage thereon shall extinguish the lien as to any installments of such assessments levied prior to the date of sale. A Mortgage or other purchaser of a Parcel or Condominium Unit who acquires title pursuant to such a foreclosure shall not be personally liable for assessments levied on such Parcel or Condominium Unit prior to the date of such acquisition of title; provided that such unpaid assessments shall be deemed to be Common Expenses collectible from Owners of all Parcels and Condominium Units subject to assessment under Section 11.7, including the Person acquiring such title and its successors and assigns.

#### 11.9 Failure to Assess.

Any failure of the Board to fix assessment amounts or rates or to deliver or mail assessment notices shall not be deemed a waiver or modification of the right to impose assessments or a release of any Owner from the obligation to pay assessments. In such event, each Owner shall continue to pay Base Assessments on the same basis as for the most recent year for which an assessment was expressly levied until a new assessment is levied, at which time the Association may retroactively assess any shortfalls in collections.

# 11.10 Exempt Property.

The Common Area, all Limited Common Area, if any, and any property that has been dedicated to and accepted by any governmental authority or public utility (including, without limitation, by deed, easement or license) or any property over which any governmental authority or public utility has control by way of a lease, license agreement or similar instrument shall be exempt from payment of Base Assessments and Special Assessments.

# 11.11 Reallocation Upon Annexation or Withdrawal of Property.

If property is annexed to or withdrawn from River Bend, the Association shall, within sixty (60) days after the date of annexation or withdrawal, recompute the budget for the fiscal year in question and recompute all applicable assessments for each Parcel and Condominium Unit. Newly annexed Parcels and Condominium Units shall be subject to

Page 27 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

assessment in accordance with Section 11.7. The Association shall send notices of any assessment applicable to the newly-annexed Parcels or Condominium Units and of any adjustments to assessments applicable to Parcels and Condominium Units which were within River Bend prior to the date of annexation or withdrawal within sixty (60) days after the date of annexation or withdrawal. Assessments under this Section 11.11 shall be due and payable on or before a date set forth in the notice, which date shall be not less than thirty (30) days after the date the notice is mailed. To the extent that any adjustment pursuant to this Section 11.11 results in a credit for assessments payable with respect to a Parcel or Condominium Unit, such credit shall be applied toward the next due payment or payments of the applicable assessment.

## 11.12 Default in Payment of Assessments; Enforcement of Lien.

If an assessment or other charge levied pursuant to this Section 11 is not paid within thirty (30) days after its due date, such assessment or charge shall become delinquent and shall bear interest from the due date until paid at the rate set forth in Section 11.13. In addition, the Association may: (i) suspend the delinquent Owner's voting rights, if any, and right of such Owner and such Owner's family members, guests, invitees, employees, tenants, and licensees to use any recreational facilities within the Common Area or any Limited Common Area; (ii) declare all remaining periodic installments of any annual assessment or any other amounts owed by such Owner to the Association immediately due and payable; (iii) file a statement of lien against such Owner's Parcel or Condominium Unit and foreclose the lien in accordance with Section 11.8; (iv) bring an action to recover monetary damages; and/or (v) exercise any other right or remedy available to it at law, in equity, or under the River Bend Documents.

#### 11.13 Interest and Late Charges.

Any assessment or other charge which is delinquent pursuant to Section 11.12 shall bear interest from the due date until paid at a rate three percentage points per annum above the prime rate, as of such due date, announced by such financial institution operating in Oregon as selected by the Board or at such other rate as may be established by the Board from time to time, or, if less, at the maximum rate which may be charged under applicable law. In addition, a late charge may be imposed for each delinquent assessment in an amount established from time to time by the Board; provided that such late charge shall not exceed 30% of the delinquent amount.

#### 12. Enforcement of Covenants.

#### 12.1 Remedies.

In the event of any breach of or other non-compliance with any provision of the River Bend Documents (other than the provisions of Section 11, as to which the rights and remedies set forth therein shall apply), the Association may: (i) bring an action to recover monetary damages; (ii) institute a proceeding in equity to obtain injunctive or other equitable relief; (iii) suspend the breaching or non-complying Owner's voting rights, if any, and right of such Owner and such Owner's family members, guests, invitees, employees, tenants, and licensees to use any recreational facilities within the Common Area or any Limited Common

Page 28 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

Area; (iv) exclude such Owner's contractors, subcontractors, agents, employees, or other invitees from the Properties (in the case of a violation of Section 6 or 7); (v) impose reasonable fines against such Owner in such amount as the Board deems appropriate in response to the violation; (vi) enter the Parcel or Condominium Unit in question, remove, abate, modify, or replace the item which is the cause of such violation in a manner that results in conformance with the River Bend Documents, and assess the cost thereof against the such Parcel or Condominium Unit and collect the same as a Specific Assessment, unless otherwise prohibited under this Declaration; and/or (vii) exercise any other right or remedy available to it at law, in equity, or under the River Bend Documents.

#### 12.2 Rights of Owners.

Any action to enforce the River Bend Documents may be instituted by Declarant, by the Board, or by any property manager retained by the Board, in each case in the name and on behalf of the Association. If, after written request from an aggrieved Owner, none of the foregoing Persons commences an action to enforce the River Bend Documents within a reasonable period, then the aggrieved Owner may bring such an action independently.

# 12.3 Limitations on Liability.

Reasonable and good faith exercise of any rights of entry set forth in the River Bend Documents shall not subject Declarant, the Association, the Board, the Design Review Committee, any other committee established under the Bylaws, or their respective partners, officers, directors, members, managers, agents, employees, or contractors to any liability for trespass, conversion, or other claim for damages. Neither Declarant, the Association, the Board, the Design Review Committee, any other committee established under the Bylaws, nor their respective partners, officers, directors, members, managers, agents, employees, or contractors shall be liable to any Owner or other Person for failure at any time to enforce any of the River Bend Documents.

# 12.4 Recovery of Costs and Fees.

In the event any suit, action, or other proceeding is instituted to enforce any of the River Bend Documents or in connection with any dispute arising thereunder, the prevailing party shall be entitled to recover its costs and expenses incurred in connection therewith, including such amount as the court may determine to be reasonable as attorneys' fees at trial and on any appeal or review.

#### 12.5 Remedies Not Exclusive.

An election to pursue any remedy provided for violation of the River Bend Documents shall not prevent concurrent or subsequent exercise of other rights or remedies permitted thereunder. The remedies provided in this Declaration and the other River Bend Documents are not exclusive, but shall be in addition to all other remedies, including actions for damages and suits for injunctions and specific performance, available under at law or in equity.

# 13. Maintenance of Properties.

#### 13.1 Association's Responsibility.

- Subject to any delegation of responsibility to a Sub-Association 13.1.1 pursuant to a Supplemental Declaration, the Association shall maintain (which, for all purposes of this Section 13, shall include repair and replacement as needed) and keep in good repair the Area of Common Responsibility, including: (i) all landscaping and other flora, parks, signage, structures, private streets, tennis courts, swimming pools, water features, fencing, and bicycle and pedestrian pathways or trails situated in the Area of Common Responsibility; (ii) landscaping, sidewalks, street lights, and signage within the public rights-of-way within or abutting the Properties, and landscaping and other flora within any public utility easements and conservation easements within the Properties (subject to the terms of any easement agreement relating thereto); provided, however, that each Owner shall landscape, irrigate, and maintain any area within public rights-of-way between such Owner's Parcel or Condominium Unit and the paved roadway located in rights-of-way adjacent to such Parcel or Condominium Unit after the date a certificate of occupancy is issued with respect to Improvements on such Parcel or with respect to such Condominium Unit; and (iii) any property or facilities owned by Declarant and made available, on a temporary or permanent basis, for the primary use and enjoyment of the Association and its Members, which property and facilities shall be identified by written notice from Declarant to the Association and remain a part of the Area of Common Responsibility until such time as Declarant revokes such privilege of use and enjoyment by written notice to the Association.
- 13.1.2 Easements are hereby reserved over the Properties to the extent necessary to enable the Association to fulfill its maintenance responsibilities pursuant to this Section 13.1.
- 13.1.3 The Association may maintain property which it does not own, including publicly owned property, conservation easements held by nonprofit entities, and other property dedicated to public use, if the Board determines that such maintenance is in the interests of the Association and the Members.
- associated with maintenance of the Area of Common Responsibility shall be Common Expenses to be included in the Base Assessment, without prejudice to the right of the Association to seek reimbursement from the owner(s) of, or other Persons responsible for, certain portions of the Area of Common Responsibility pursuant to this Declaration, other recorded covenants, or agreements with such owner(s) or other Persons.

#### 13.2 Owner's Responsibility.

Each Owner shall maintain such Owner's Parcel or Condominium Unit and all structures, parking areas, landscaping, and other Improvements on or comprising such Parcel or Condominium Unit in good order and repair and in a manner consistent with all applicable covenants and legal requirements, unless such maintenance responsibility is assumed by or

Page 30 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

assigned to the Association pursuant to any Supplemental Declaration. In addition to any other enforcement rights provided in the River Bend Documents, if an Owner fails properly to perform the foregoing maintenance responsibilities or the maintenance responsibilities set forth in the proviso to clause (ii) of Section 13.1.1, the Association may enter onto the property in question and perform such maintenance responsibilities and assess all costs incurred by the Association against the Parcel or Condominium Unit and the Owner thereof as a Specific Assessment in accordance with Section 11.5.

# 14. Insurance and Casualty Losses.

# 14.1 Association Insurance.

- 14.1.1 The Association shall obtain blanket "all-risk" property insurance, if available at reasonable cost, for all insurable Improvements in the Area of Common Responsibility. If blanket "all-risk" coverage is not available at reasonable cost, the Association shall obtain fire and extended coverage insurance, including coverage for vandalism and malicious mischief. The Association may also obtain flood insurance, if available at reasonable cost. The face amount of any policy shall be sufficient to cover the full replacement cost of the insured property.
- the Association and its Members for damage or injury caused by the negligence of the Association or any of its Members, employees, agents, or contractors while acting on its behalf. If available at reasonable cost, such policy shall have at least a \$1,000,000 combined single limit with respect to bodily injury and property damage and at least a \$3,000,000 limit per occurrence and in the aggregate. To the extent available at reasonable cost, such public liability insurance shall also include protection against water damage liability, liability for non-owned and hired automobiles, liability for property of others, and, if applicable, elevator collision, garage-keeper's liability, host liquor liability, contractual and all-written contract insurance, employers' liability insurance, and coverage against such other risks as are customarily covered with respect to projects similar in construction, location, and use to River Bend.
- 14.1.3 Each policy of insurance contemplated by this Section 14.1 may contain a reasonable deductible which shall not be subtracted from the face amount of the policy in determining whether the insurance provides the required coverage. In the event of an insured loss, the deductible shall constitute a Common Expense and be included in the Base Assessments.
- 14.1.4 All insurance coverage obtained by the Association shall: (i) be written with a company authorized to issue insurance in Oregon which holds a Best's rating of "A" or better and is assigned a financial size category of IX or larger as established by A.M. Best Company, Inc., if available at reasonable cost, or, if not so available, with a company authorized to issue insurance in Oregon which holds the most nearly equivalent rating which is so available; (ii) be written in the name of the Association as trustee for the benefited parties and include a standard mortgagee's clause; (iii) vest in the Board exclusive authority to adjust losses; provided, however, that no Mortgagee having an interest in such losses shall be prohibited from

Page 31 – AMENDED AND RESTATED MASTER DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR RIVER BEND

participating in the settlement negotiations, if any, related to the loss; (iv) not be brought into contribution with insurance purchased by individual Owners, occupants, or their Mortgagees; and (v) have an inflation guard enforcement, if available at reasonable cost. If a policy contains a co-insurance clause, it shall also have an agreed amount endorsement. The Association shall arrange for an annual review of the sufficiency of insurance coverage by one or more qualified Persons, at least one of whom shall be in the real estate industry and familiar with construction in the City.

- 14.1.5 The Board shall use reasonable efforts to secure insurance policies containing endorsements that: (i) waive subrogation as to any claims against the Board and the Association's officers, employees, and property manager (if any), as well as the Owners and their tenants, servants, agents, and guests; (ii) waive the insurer's rights to repair and reconstruct instead of paying cash; (iii) preclude cancellation, invalidation, suspension, or non-renewal by the insurer on account of any one or more individual Owners, or on account of any curable defect or violation without prior written demand to the Association to cure the defect or violation and allowance of a reasonable time to cure; (iv) exclude individual Owners' policies from consideration under any "other insurance" clause; and (v) require at least thirty (30) days written notice to the Association (and any Mortgagee or other insureds named therein) prior to any cancellation, substantial modification, or nonrenewal.
- 14.1.6 The Association shall obtain worker's compensation insurance and employer's liability insurance, if and to the extent required by law, and shall use reasonable efforts to obtain errors and omission insurance coverage for officers, directors and committee members of the Association.
- 14.1.7 The Association shall obtain a fidelity bond or bonds, if available at reasonable cost, covering all Persons responsible for handling Association funds. The Board shall determine the amount of fidelity coverage in its best business judgment. Bonds shall contain a waiver of all defenses based upon the exclusion of Persons serving without compensation and shall require at least thirty (30) days written notice to the Association prior to any cancellation, substantial modification, or nonrenewal.
- 14.1.8 The premiums for all insurance and bonds contemplated by this Section 14.1 shall constitute Common Expenses and be included in the Base Assessments.

# 14.2 Owners' Insurance.

14.2.1 By accepting a deed to or entering into a recorded contract of sale for a Parcel or a Condominium Unit, each Owner covenants and agrees to carry, at such Owner's expense, blanket "all-risk" property insurance on the structures on or comprising such Parcel or Condominium Unit, providing full replacement cost coverage less a reasonable deductible. In addition, each Owner may obtain such other and additional insurance coverage on and in relation to such Owner's Parcel or Condominium Unit as such Owner concludes to be desirable; provided, however, that none of such insurance coverage shall affect any insurance coverages obtained by the Association or cause a diminution or termination of any such coverage. Any

insurance obtained by an Owner shall include a waiver of the insurer's right of subrogation against the Association and other Owners.

14.2.2 In the event of damage to or destruction of the structures on or comprising a Parcel or Condominium Unit, the Owner thereof shall proceed promptly to repair or reconstruct such structures in a manner consistent with the original construction or as may otherwise be approved pursuant to Section 6. Alternatively, the Owner of a Parcel shall clear such Parcel of all debris and ruins and maintain such Parcel in a neat and attractive, landscaped condition consistent with the River Bend Documents. The Owner shall pay any costs of repair or replacement that are not covered by insurance proceeds. If the required repair or restoration is not commenced within 180 days after the date of damage or destruction, or if repair and reconstruction is commenced but is not diligently pursued or is abandoned for a period of more than ninety (90) days, the Association may, following reasonable notice, impose a reasonable fine on the Owner until repair and reconstruction is commenced, unless the Owner can demonstrate to the satisfaction of the Association that such failure is due to circumstances beyond the Owner's reasonable control.

#### 14.3 Damage and Destruction.

- 14.3.1 Promptly after any damage to or destruction of all or any part of the Properties covered by insurance written in the name of the Association, the Board or its duly authorized agent shall file and adjust all insurance claims and obtain reliable and detailed estimates of the cost of repair or reconstruction. Repair or reconstruction, as used in this Section 14.3, means repairing or restoring the property to substantially the condition in which it existed prior to the damage or destruction, taking into account changes or Improvements necessitated by changes in applicable legal requirements.
- 14.3.2 Any damage to or destruction of the Common Area shall be repaired or reconstructed unless Class A Members together holding at least 75% of the total votes held by Class A Members and Declarant (so long as Declarant owns any property in the First Phase or any Potential Expansion Property) decide within sixty (60) days after the loss not to repair or reconstruct. If either the insurance proceeds or reliable and detailed estimates of the cost of repair or reconstruction, or both, are not available to the Association within such 60-day period, then the period shall be extended until such funds and such information are available. Assessments of the Association shall not be abated during the period of insurance adjustment and repair and reconstruction.
- 14.3.3 If, pursuant to Section 14.3.2, damage to or destruction of the Common Area is not to be repaired or reconstructed and no alternative Improvements are authorized, the affected property shall be cleared of all debris and ruins and maintained by the Association in a neat and attractive, landscaped condition consistent with the River Bend Documents. In such event, any remaining insurance proceeds shall be retained by the Association and used for such purposes as the Board may determine.

# 14.4 Association as Attorney-in-Fact.

By accepting a deed to or entering into a recorded contract of sale for a Parcel or a Condominium Unit, each Owner irrevocably constitutes and appoints the Association as such Owner's true and lawful attorney-in-fact in such Owner's name, place, and stead for the purpose of dealing with the Improvements in the Area of Common Responsibility upon damage or destruction as provided in this Section 14 or upon a complete or partial Condemnation as provided in Section 15. As attorney-in-fact, the Association shall have full and complete authority, right, and power to make, execute, and deliver any contract, assignment, deed, waiver, or other instrument with respect to the interest of any Owner which may be necessary or appropriate to exercise the foregoing powers.

#### 15. Condemnation.

#### 15.1 Partial Condemnation.

If any part of the Common Area is Condemned, the resulting award shall be payable to the Association as trustee for all Owners and shall be applied as follows:

15.1.1 If the Condemnation involves a portion of the Common Area on which Improvements have been constructed, the Association shall restore or replace such Improvements on the remaining land included in the Common Area, to the extent available, unless, within sixty (60) days after the date of such Condemnation, Class A Members together holding at least 75% of the total votes held by Class A Members and Declarant (so long as Declarant owns any property in the First Phase or any Potential Expansion Property) otherwise decide.

15.1.2 If the Condemnation does not involve any Improvements on the Common Area, or if a decision is made pursuant to Section 15.1. not to repair or restore, or if funds from the Condemnation award remain after the completion of such restoration or replacement, then such award or such remaining funds shall be retained by the Association and used for such purposes as the Board may determine.

# 15.2 Complete Condemnation.

If all of River Bend is Condemned, the regime created by this Declaration shall terminate and the portion of the Condemnation award attributable to the Common Area shall be distributed in proportion to the respective obligations for Base Assessments for the fiscal year during which the Condemnation occurs, first to the Mortgagees and then to the Owners, as their interests may appear.

# 16. Mortgage Provisions.

#### 16.1 Parties Benefited.

The provisions of this Section 16 are for the benefit of holders of each First Mortgage on Condominium Units, and shall apply to all of the River Bend Documents, notwithstanding any provision to the contrary contained therein.

#### 16.2 Notices.

Any holder of a First Mortgage on a Condominium Unit who provides a written request to the Association stating the name and address of such holder and the street address of the Condominium Unit to which its Mortgage relates (an "Eligible Holder"), shall be entitled to timely written notice from the Association of: (i) any Condemnation or casualty loss which affects the Condominium Unit which is subject to the Mortgage held by such Eligible Holder; (ii) any delinquency in the payment of assessments or charges attributable to such Condominium Unit, if such delinquency has continued for a period of sixty (60) days, or any other violation of the River Bend Documents relating to such Condominium Unit or the Owner or occupant thereof which is not cured within sixty (60) days; (iii) any lapse, cancellation, or material modification of any insurance policy maintained by the Association; and (iv) any proposed action which would require the consent of a specified percentage of Mortgagees.

#### 16.3 Special FNMA/FMC Provision.

For purposes of satisfying requirements relating to the Common Area imposed by the Federal National Mortgage Association or the Freddie Mac Corporation, as amended from time to time, the following provisions shall apply in addition to and not in lieu of any other provision of this Declaration:

Without the prior approval in writing by at least a 67% vote 16.3.1 of the total allocated votes of the Association and by at least 51% of the Eligible Holders, the Association shall not: (i) amend the Condominium Documents to allow an increase of the amount of any assessments that may be levied against a Parcel or Condominium Unit by more than 25% over the previously assessed amount; (ii) change the method of determining the obligations, assessments, dues, or other charges that may be levied against a Condominium Unit; provided that a decision by the Board or provisions of any Supplemental Declaration shall not be violative of this clause if such decision or Supplemental Declaration is otherwise authorized by this Declaration; (iii) by act or omission change, waive, or abandon any scheme of regulations or enforcement pertaining to maintenance of Improvements and the Common Area; provided that the issuance and amendment of Design Guidelines or other rules, regulations, guidelines, or restrictions in accordance with the River Bend Documents shall not constitute a change, waiver, or abandonment within the meaning of this clause; nor (iv) use casualty insurance proceeds resulting from damage to or destruction of the Common Area for any purpose other than the repair, replacement, or reconstruction of such property, except to the extent expressly set forth in this Declaration.

16.3.2 In addition, unless at least 67% of the Eligible Holders or 67% of the Condominium Class A Members have given their prior approval in writing, the Association shall not, except to the extent permitted by applicable law, by act or omission seek to abandon, partition, subdivide, encumber, sell, or transfer all or any portion of the real property comprising the Common Area that the Association owns, directly or indirectly; provided that the granting of easements for public utilities or other similar purposes consistent with the intended use of the Common Area shall not be deemed a transfer within the meaning of this clause

#### 16.4 No Priority.

No provision of the River Bend Documents shall be construed to give any Owner or other Person priority over any rights of the holder, insurer, or guarantor of any First Mortgage in the case of a distribution to such Owner or other Person of insurance proceeds or Condemnation awards for damage to or destruction or a Condemnation of the Common Area.

#### 16.5 Notice to Association.

Upon request, each Owner shall furnish to the Association the name and address of the holder of any Mortgage encumbering such Owner's Condominium Unit.

## 16.6 Amendment by Board.

In the event that the Federal National Mortgage Association and the Freddie Mac Corporation subsequently eliminates any of their respective requirements which necessitate the provisions of this Section 16 or makes any such requirements less stringent, the Board may amend this Section 16 to reflect such changes, without approval of the Owners.

#### 16.7 Failure of Mortgagee to Respond.

Any Mortgagee who receives a written request from the Board to consent to or approve any proposed action shall be deemed to have consented to and approved such action if the Association does not receive a written response from the Mortgagee within 30 days after the date of the Association's request.

# 17. Declarant's Rights.

# 17.1 Transfer of Rights.

Any or all of the special rights and obligations of Declarant set forth in this Declaration or in the other River Bend Documents may be transferred to other Persons, without the consent of any Mortgagee or Owner, so long as the transfer does not enlarge any such right or reduce any such obligation. No such transfer shall be effective unless it is reflected in a written instrument signed by Declarant and recorded in the County Records. Nothing in this Declaration shall be construed to require Declarant or any successor to develop any of the Potential Expansion Property in any manner whatsoever or to annex any such property to River Bend.

#### 17.2 Construction and Sales Period.

So long as construction on and initial sales of Parcels continue, Declarant may maintain and carry on upon portions of the Common Area and the Limited Common Area, if any, such facilities and activities as are, in the sole opinion of Declarant, reasonably required, convenient, or incidental to such construction or sales, including business offices, signs, model units, and sales offices. Declarant hereby reserves easements for access to and use of such facilities. Declarant's unilateral right to use the Common Area and the Limited Common Area, if any, for the purposes set forth in this Section 17.2 shall not be exclusive and shall not interfere unreasonably and substantially with use of the Common Area and the Limited Common Area, if any, by Owners, except to the extent, if any, that portions of the Common Area or the Limited Common Area, if any, are leased to Declarant for a reasonable rent.

#### 17.3 Other Covenants.

No Person shall record any declaration of covenants, conditions, and restrictions, any condominium declaration, or any similar instrument affecting any portion of the Properties without Declarant's prior written consent. Any such instrument which is recorded without such consent shall be void and of no force and effect unless subsequently approved by Declarant.

#### 17.4 Amendment.

Notwithstanding any other provision of the River Bend Documents, this Section 17 may not be amended without the prior written consent of Declarant.

## 17.5 Termination of Special Rights.

The rights contained in this Section 17 shall terminate upon the earlier of:
(i) forty (40) years after the date this Declaration is recorded, or (ii) the recording by Declarant of a written statement to the effect that all initial sales activity in connection with River Bend has ceased. Thereafter, Declarant may continue to use the Common Area and the Limited Common Area, if any, for purposes stated in this Section 17 only pursuant to a lease agreement between Declarant and the Association or the appropriate Sub-Association which provides for rental payments based on the fair market rental value of such portion of the Common Area or the Limited Common Area, if any.

#### 18. General Provisions.

#### 18.1 <u>Term</u>.

This Declaration shall run with and bind the Properties, and shall inure to the benefit of and shall be enforceable as provided herein by the Association, any Owner, and their respective legal representatives, heirs, successors, and assigns, for a term of forty (40) years from the date this Declaration is recorded. After such period, this Declaration shall automatically be extended for successive periods of ten (10) years each, unless a written instrument, signed by Owners together holding a majority of the votes held by Class A Members and by Declarant

(prior to Turnover), is recorded in the County Records within one year prior to the end of any such extension period providing for the termination of this Declaration, in which case this Declaration shall terminate as provided therein.

#### 18.2 Amendment.

#### 18.2.1 By Declarant.

At any time and from time to time prior to the initial conveyance of a Parcel to an Owner other than Declarant, Declarant may unilaterally amend this Declaration. Thereafter, until the termination of the Class B membership pursuant to Section 9.3.2, Declarant may, subject to the provisions of Section 16, unilaterally amend this Declaration; provided, however, that no such amendment shall materially and adversely affect any material rights of any Members hereunder without the consent of affected Members who together hold a majority of the total votes held by all affected Members.

# 18.2.2 By Owners.

In addition to the rights of Declarant pursuant to Section 18.2.1, this Declaration may, subject to the provisions of Section 16, be amended at any time and from time to time upon the affirmative vote or written consent, or any combination thereof, of: (i) Class A Members together holding at least 75% of the total votes held by Class A Members, and (ii) Declarant, so long as Declarant owns any of the property in the First Phase or any of the Potential Expansion Property.

# 18.2.3 Validity and Effective Date of Amendments.

Amendments to this Declaration shall become effective upon recordation in the County Records, unless a later effective date is specified therein. Any procedural challenge to an amendment shall be made within one year after the date of its recordation or such amendment shall be conclusively presumed to have been validly adopted. In no event shall a change of conditions or circumstances operate to amend any provision of this Declaration.

# 18.2.4 Declarant Rights.

No amendment to this Declaration or any of the other River Bend Documents may remove, revoke, or modify any right or privilege of Declarant hereunder or under any of the other River Bend Documents without the written consent of Declarant and any assignee of such right or privilege.

#### 18.2.5 Boundaries.

No amendment to this Declaration shall change the boundaries of any Parcel or Condominium Unit without the consent of the affected Owner.

#### 18.3 Severability.

Invalidation of any provision of this Declaration, in whole or in part, or of any application of a provision of this Declaration, by judgment or court order shall in no way affect other provisions or applications.

#### 18.4 Notice of Sale or Transfer of Title.

Any Owner wishing to sell or otherwise transfer title to such Owner's Parcel or Condominium Unit (other than by operation of law) shall give the Board at least seven days prior written notice of the name and address of the purchaser or transferee, the anticipated date of the transfer of title, and such other information as the Board may reasonably specify from time to time. The transferor shall continue to be jointly and severally responsible with the transferee for all obligations of the Owner of the Parcel or Condominium Unit, including assessment obligations, until the date on which such notice is received by the Board, notwithstanding the transfer of title.

#### 18.5 Revocation.

Except as provided in Section 15.2, this Declaration shall not be revoked without the consent of all of the Members in a written instrument recorded in the County Records.

#### 18.6 Notices.

All notices pursuant to this Declaration shall be in writing. Notice to any Member shall be considered delivered and effective upon personal delivery, or three days after deposit in the U.S. Mail, certified, return receipt requested, addressed to the address of such Member on file in the records of the Association at the time of such mailing. Notice to the Board, the Association, or the Design Review Committee shall be considered delivered and effective upon personal delivery, or three days after deposit in the U.S. Mail, certified, return receipt requested, addressed to the Association, the Board, or the Committee at such address as may be established by the Association from time to time by notice to the Members. General notices to all Members need not be certified, but may be sent by regular first class mail.

#### 18.7 Waiver.

No failure on the part of Declarant, the Association, the Board, or the Committee to give notice of default or to exercise any right or remedy, nor any delay in exercising any right or remedy, shall operate as a waiver, except as specifically provided herein in the event the Board or the Committee fails to respond to certain requests. No waiver shall be effective against Declarant, the Association, the Board, or the Committee unless it is in writing, signed, as applicable, by Declarant, by the President or Vice President of the Association on behalf of the Association or the Board, or by the chairman of the Committee on behalf of the Committee.

#### 18.8 Conflicts Between Documents.

In case of any conflict between this Declaration and the Articles of Incorporation or the Bylaws, this Declaration shall control. In case of conflict between the Articles of Incorporation and the Bylaws, the Articles of Incorporation shall control. In case of conflict between this Declaration and the Design Guidelines or the Use Guidelines and Restrictions, this Declaration shall control.

IN WITNESS WHEREOF, Declarant has executed this Declaration effective as of the date set forth above.

RIVER BEND LIMITED PARTNERSHIP, an Oregon limited partnership

By: The Bend Company, an Oregon corporation, General Partner

By: William L. Smith, Président

STATE OF OREGON ) ss.
County of Deschutes )

The foregoing instrument was acknowledged before me on this <u>2nd</u> day of <u>June</u>, 1998, by William L. Smith, who is President of the Bend Company, an Oregon corporation, general partner of RIVER BEND LIMITED PARTNERSHIP, an Oregon limited partnership, on behalf of the limited partnership.

OFFICIAL SEAL
REX ANN EVERETT
NCTARY PUBLIC-OREGON
COMMISSION NO. 055180
MY COMMISSION EXPIRES JUNE 19, 2000

Notary Public for Oregon

My Commission Expires: June 19, 2000